Author’s response to reviews

Title: Development of a brief learning environment measure for use in healthcare professions education: the Healthcare Education Micro Learning Environment Measure (HEMLEM)

Authors:

Rachel Isba (rachel.isba@lancaster.ac.uk)

Christiana Rousseva (christiana.rousseva@googlemail.com)

Katherine Woolf (k.woolf@ucl.ac.uk)

Lucie Byrne-Davis (lucie.byrne-davis@manchester.ac.uk)

Version: 3 Date: 28 Feb 2020

Author’s response to reviews:

As with all of our resubmissions in response to reviewers, we have included a letter in the supplementary materials section.